REVIEW HISTORY

Manuscript Title: Experience-based considerations for investigating complex and evolving issues using a longitudinal mixed methods approach

Submitted: 05 July, 2024 Accepted: 10 January, 2025 Published: 25 January, 2025

Corresponding Author: "Antonella Maria Giacosa"

Email: antonellamaria.giacosa@unito.it

Doi: 10.19044/llc.v12no1a1

PEER REVIEW

Reviewer 1: Blinded

Reviewer 2: Andrew Nelson